Author's response to reviews

Title: Simple technique for evacuation of traumatic subcutaneous haematomas under tension.

Authors:

George MR. Chami (georgechami@hotmail.com)
Edward DR. Hatley (edwardhatley@yahoo.co.uk)
Belinda DR. Chami (belindachami@hotmail.com)
Hossam MR. Dabis (h.dabis@btopenworld.com)

Version: 5 Date: 7 December 2005

Author’s response to reviews:

Dear Sir or Madam,

The authors would be grateful if you would consider the revised manuscript of the above article for publication in the BMC Emergency Medicine. The authors have all read the manuscript and agree to its submission. It is an original piece of research and is not under consideration at any other journal and has not been previously published.

Please note that no further changes has been made to the manuscript since the last revision submission.

We hope you find it acceptable.

Kind Regards

George Chami